**Author’s response to reviews**

**Title:** Modeling technology innovation: How science, engineering, and industry methods can combine to generate beneficial socioeconomic impacts

**Authors:**

Vathsala I Stone (vstone@buffalo.edu)
Joseph P Lane (joelane@buffalo.edu)

**Version:** 7  **Date:** 4 May 2012

**Author’s response to reviews:**

All comments have been addressed and all copy edits have been accepted. This includes revisions to the Figures as well as the addition of all the complete Figure Legends.

There were 2 revisions made to the cover page which have been made by our office. These two revisions included the correct name of the University to read, University at Buffalo (SUNY) and not as revised by the copy editor. We also had to correct the initials of one of the authors. Both changes have been accepted to the paper.

This latest upload is a clean version of our manuscript, including a clean version of all figures.

Thank you for your assistance.